Author's response to reviews

Title: An international comparative study of blood pressure in populations of European vs. African descent

Authors:

Richard S Cooper (rcooper@lumc.edu)
Katharina Wolf-Maier (kwolf3@lumc.edu)
Amy Luke (aluke@lumc.edu)
Adebowale Adeyemo (deboadeyemo@scientist.com)
Jose R Banegas (joseramon.banegas@uam.es)
Terrence Forrester (terrence.forrester@uwimona.edu.jm)
Simona Giampaoli (simona.giampaoli@iss.it)
Michel Joffres (michel.joffres@dal.ca)
Mika Kastarinen (Mika.Kastarinen@uku.fi)
PaolaPrimatesta (p.primatesta@public-health.ucl.ac.uk)
Birgitta Stegmayr (Birgitta.Stegmayr@medicin.umu.se)
Michael Thamm (thammm@rki.de)

Version: 3 Date: 15 November 2004

Author’s response to reviews:

Dear Ms. Puebla,

Please find attached a revised version of the manuscript and figures for 1684190564428749. The following formatting revisions were made as per your request.

1) Unnecessary capitalization was removed from the title in both places.
2) Country was added to all affiliations.
3) Keywords section was removed from the manuscript.
4) Corresponding author details were moved to the first page. (Double spacing had to be changed to single in order to accomodate this.)
5) Abstract was put on the second page.
6) Introduction was renamed Background.
7) Square brackets were used for references instead of parentheses.
8) Competing interests statement was changed.
9) Funding info was placed in acknowledgements section.
10) Tables was given a solid black border.
11) Figures were cropped according to your directions.

Thank you.

Richard Cooper, MD